Author's response to reviews

Title: Improvised peritoneal dialysis in an 18 month child with severe malnutrition (kwashiorkor) and acute kidney injury: a case report

Authors:

Francis Fredrick (fredrick.francis78@gmail.com)
Gudila Valentine (gudilav@gmail.com)

Version: 2 Date: 12 February 2013

Author's response to reviews:

Dear Editors,

Thank you for your prompt reply. We have made changes in the manuscript as recommended. We have included ethnicity of our patient and the gender of the patient has been identified with the word male as recommended. We have also included the phrase stating that this is the first case of peritoneal dialysis in a child with severe malnutrition in the introduction section.

We are forwarding our sincere thanks in advance.

Yours cordially,

Francis Fredrick.